Author's response to reviews

Title: Phenolic content and antioxidant property of the bark extracts of Ziziphus mucronata Wild. subsp. mucronata Wild.

Authors:

Olufunmiso O Olajuyigbe (funmijuyigbe12@yahoo.com)
Anthony J Afolayan (Afolayan@ufh.ac.za)

Version: 2 Date: 13 July 2011

Author's response to reviews: see over
The Editor  
BMC Complementary and Alternative Medicine  

Dear sir/madam,  

Re: SUBMISSION OF MANUSCRIPT  

In response to editorial comments on our article: MS: 1601742945428983 - Phenolic content and antioxidant property of the bark extracts of Ziziphus mucronata Wild. subsp. mucronata Wild, major revisions have been done.  

These revisions included:  
   1. Responding to the suggestions of each reviewer’s reports.  
   2. Repeating DPPH and ABTS experiment to determine the IC$_{50}$ of each extract.  
   3. Doing major review, editing and correcting the manuscript.  

All changes made during the editing and reviewing of the manuscript are indicated in red letters on the revised version of the article.  

I hereby resubmit our article for further review and eventual publication in your journal.  

I also declare that the article has not been submitted and not under consideration for publication elsewhere. If you eventually accept the article for publication in your journal, the same document will not be submitted to any other journal for publication.  

Kindly acknowledge receipt.  

Regards,  

Prof AJ Afolayan